Author’s response to reviews

Title: A novel Cys328-terminator mutant implicated in severe coagulation factor XIII deficiency: A case report

Authors:
Ruimin Cai (ymxcrm@163.com)
Yi Li (candiceliyi@126.com)
Wenyang Wang (wenyang11@sina.com)
Qiang Feng (fengqiang915@163.com)

Version: 2 Date: 24 Jun 2020

Author’s response to reviews:

Response to the comments
Reviewer 1:
Thank you for your comments, I must point out the fact that the patient in our report didn't be diagnosed with coagulation factor XIII deficiency, while given the symptomatic treatment everytime. She was diagnosed with coagulation factor XIII deficiency at Tianjin Blood Institute at the age of 20, soon after, she came to our hospital for the symptomatic treatment, we went through her medical records and carried out the clot solubility test to verify her diagnosis and decided to do the genetic analysis.

Reviewer 2:
(1): Thank you for your comments, we will conduct the English proof-reading in our manuscript.
(2): Owing to the mutation in our report is a nonsense mutation, it cannot be predicted by the polyphen 2, but it can be predicted by mutation taster, I will point it out in the manuscript.
(3): We will enclose the figure legend of Fig 3.
(4): This is also our concern, we have tried our best to persuade the parents to conduct the genetic analysis, disappointingly, they rejected our request because of the privacy.